Author's response to reviews

Title: Transversus abdominis-plane block versus local anesthetic infiltration in lower abdominal surgery: a meta-analysis of randomized controlled trials

Authors:

Nanze Yu (yunanze@hotmail.com)
Xiao Long (pumclongxiao@126.com)
Jorge R Lujan-Hernandez (Jorge.lujan87@hotmail.com)
Julien Succar (julas19@gmail.com)
Xin Xin (xinxin.harvard@gmail.com)
Xiaojun Wang (xjwang100@hotmail.com)

Version: 4 Date: 28 May 2014

Author's response to reviews:

Thank you for your letter and the reviewers’ comments concerning our manuscript entitled “Transversus Abdominis Plane Block or Local Anesthetic Infiltration? A Meta-Analysis of Randomized Controlled Trials” (MS: 5874865781248411). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. Below is the point-by-point response. We would be grateful if you would consider our manuscript for publication.

Sincerely,

Xiaojun Wang, M.D.
xjwang100@hotmail.com
Division of Plastic and Reconstructive Surgery
Peking Union Medical College Hospital
No. 1 Shuaifuyuan, Dongcheng District Beijing 100032, China